Author's response to reviews

Title: A modified root reinforcement technique for acute aortic dissection with a weakened aortic root: a modified Florida sleeve technique and two cases report.

Authors:

Woon Heo (loofni@naver.com)
Ho-Ki Min (minhoki@naver.com)
Do Kyun Kang (med7695@naver.com)
Hee Jae Jun (cs523@inje.ac.kr)
Youn-Ho Hwang (hwangyh@paik.ac.kr)
Jin Ho Choi (jinhc72@hanmail.net)
Jin Hong Wi (wiccy@hanmail.net)

Version: 2 Date: 17 July 2013

Author's response to reviews: see over
Dear Editors:

I, along with my coauthors, would like to ask you to consider the attached manuscript entitled “A modified root reinforcement technique for acute aortic dissection with a weakened aortic root: a modified Florida sleeve technique and two cases report” for publication in Journal of Cardiothoracic Surgery as a case report.

We believe that the findings of this study are relevant to the scope of your journal and will be of interest to its readership.

This manuscript has not been published or presented elsewhere in part or in entirety, and is not under consideration by another journal. All study participants provided informed consent, and the study design was approved by the appropriate ethics review boards. All the authors have approved the manuscript and agree with submission to your esteemed journal. There are no conflicts of interest to declare.

Thank you for your consideration. I look forward to hearing from you.

Sincerely,

Ho-Ki Min, M.D.
Department of Thoracic and Cardiovascular Surgery, Haeundae Paik Hospital, Inje University College of Medicine, 875 (Jwadong) Haeundae-ro, Haeundaegu, Busan, 612-030, Korea
Tel.: 82-51-797-3135, Fax: 82-51-797-3135
E-mail: minhoki@naver.com